\section*{Rules}
~


\textbf{ Class meets} at 10:45 MWF

\textbf{ Office hours:} After the class and by appointment

The class is \textbf{Addressed }to graduate and senior
undergraduate students in math, science, and engineering. Grade is
based on regular homework assignments and a course project.

\textbf{ Text:} Gilbert Strang. Introduction to Applied
Mathematics. Instructors notes (will be distributed).

\textbf{Syllabus: }

1.G. Strang: Chapters 1,2, 3

 2. Notes: Extremal problems.


 The \textbf{grade} will be based on homework scores, in-class exam, and
course project. As a rule, the homework will be assigned each
week.
